Author's response to reviews

Title: The clinical significance of serum and bronchoalveolar lavage inflammatory cytokines in patients at risk for ARDS

Authors:

- Demosthenes Bouros (bouros@med.duth.gr)
- Michael Alexandrakis (bouros@med.duth.gr)
- Katerina M Antoniou (katerinaantoniou@yahoo.gr)
- Panagiotis Agouridakis (bouros@med.duth.gr)
- Ioannis Pneumatikos (bouros@med.duth.gr)
- Stavros Anevlavis (bouros@med.duth.gr)
- Athanasia Pataka (bouros@med.duth.gr)
- George Patlakas (gpatlak@med.duth.gr)
- Nikolaos Karkavitsas (bouros@med.duth.gr)
- Despina Kyriakou (bouros@med.duth.gr)

Version: 3 Date: 2 August 2004

Author's response to reviews:

Thank you very much for your helpful suggestions. According to them us revised our MS. The changes made concern the title page (capitalisation, qualifications, affiliations, running title, email addresses) Abstract (typos), Keywords, references (brackets), Material and Methods and Results (all suggested changes made) Also, we completed the author's contributions and the reference list. Figures are sent separately.